Author’s response to reviews

Title: Gastric and duodenal metastases of malign melanoma

Authors:

Vedat Goral Prof (vegoral@hotmail.com)
Feyzullah Ucmak PhD (ucmakfeyz@hotmail.com)
Sezgin Barutcu Dr (sezginbarutcu@yahoo.com)
Serdar Yildirim Dr (yildirs@hotmail.com)
Serdar Ileri Dr (serdari@dicle.edu.tr)
Huseyn Buyukbayram Prof (husram@dicle.edu.tr)

Version: 2 Date: 10 January 2010

Author’s response to reviews:

The article is right. But there must be 4 endoscopic pictures and 4 histopathologic pictures. Others pictures is repeats. So please remove reopeat pictures. Sincerely yours.

Prof. Dr. Vedat Goral